Author's response to reviews

Title: AMPD1 regulates mTORC1-p70 S6 kinase axis in the control of insulin sensitivity in skeletal muscle

Authors:

Andreas AK Tandelilin (andreas@ri.ncvc.go.jp)
Tetsuaki Hirase (hirase@ri.ncvc.go.jp)
Athanasius W Hudoyo (athanasius_wrin@yahoo.com)
Jidong Cheng (jidongcheng36@hotmail.com)
Keiko Toyama (ktoyama2@plum.ocn.ne.jp)
Hiroko Morisaki (hirokom@ri.ncvc.go.jp)
Takayuki Morisaki (morisaki@ri.ncvc.go.jp)

Version: 2
Date: 26 December 2014

Author's response to reviews:

Possible Reviewers

Junichi Sadoshima
Rutgers New Jersey Medical School
UNITED STATES
Email: sadoshju@njms.rutgers.edu

Richard L. Sabina, Ph.D.,
Department of Biomedical Sciences
Oakland University William Beaumont School of Medicine
Office: 516 O'Dowd Hall
Phone: 248-370-2755
Email: sabina@oakland.edu

Hiroaki Masuzaki
University of the Ryukyu
Email: hiroaki@med.u-ryukyu.ac.jp